Title: T-cell-based diagnosis of tuberculosis infection in children in Lithuania: a country of high incidence despite high bacille Calmette-Guerin vaccination coverage

Authors:

- Edita Hansted (edita.hansted@gmail.com)
- Angele Andriuskeviciene (olde@one.lt)
- Raimundas Sakalauskas (raimundas.sakalauskas@kmuk.lt)
- Rimantas Kevalas (rimantas.kevalas@kmuk.lt)
- Brigita Sitkauskiene (brigita.sitkauskiene@kmuk.lt)

Version: 2 Date: 24 February 2009

Author's response to reviews:

Dear Editor,

Many thanks for your valuable comments and suggestions.

Please find enclosed the revised version of our manuscript entitled
" T-cell-based diagnosis of tuberculosis infection in children in Lithuania: a country of high incidence despite a high coverage with bacille Calmette-Guerin vaccination " (Manuscript Number: MS: 1863106154239820).

We hope that the current version is fully acceptable for publication in BMC Respiratory Medicine.

Thank you and sincerely Yours

Brigita Sitkauskiene, MD, PhD